Author’s response to reviews

Title: Combining Glycosylated Hemoglobin A1c and Fasting Plasma Glucose for Diagnosis of Type 2 Diabetes in Chinese Adults

Authors:

Miao Mo (10211020008@fudan.edu.cn)
Weijian Zhong (wzhong@scdc.sh.cn)
Genming Zhao (gmzhao@shmu.edu.cn)
Ye Ruan (yruan@scdc.sh.cn)
Hua Zhang (11211020013@fudan.edu.cn)
Liang Shi (lshi@scdc.sh.cn)
Dajiang Lu (ludajiang2000@yahoo.com.cn)
Qundi Yang (gdyang@scdc.sh.cn)
Yanyun Li (yyli@scdc.sh.cn)
Qingwu Jiang (jiangqw@fudan.edu.cn)
Rui Li (rlrli@scdc.sh.cn)
Wanghong Xu (wanghong.xu@fudan.edu.cn)

Version: 4 Date: 29 September 2013

Author’s response to reviews: see over
Eloisa Nolasco  
Journal Editorial Office  
BioMed Central  
E-mail: editorial@biomedcentral.com  
Web: http://www.biomedcentral.com

Re: MS 1251774703934367

Dear Dr. Eloisa Nolasco:

Please find attached a revised version of our manuscript “Combining Glycosylated Hemoglobin A1c and Fasting Plasma Glucose for Diagnosis of Type 2 Diabetes in Chinese Adults”, which we would like to resubmit for publication in your journal of *BMC Endocrine Disorders*.

Revisions in the text are shown in red. In accordance with reviewer Hillary Keenan’s suggestion, we adjust the spelling and reference to tables, and made some language corrections. We hope that the revisions in the manuscript will be sufficient to make our manuscript suitable for publication in *BMC Endocrine Disorders*.

We shall look forward to hearing from you at your earliest convenience.

Sincerely yours,

Wang-hong Xu, MD, PhD

Associate Professor
Department of Epidemiology
School of Public Health Fudan University
138 Yi Xue Yuan RD
Shanghai 200032, China
Tel: 86-21-54237679
E-mail: wanghaihong.xu@fudan.edu.cn